Author’s response to reviews

Title: Knowledge, Attitudes, and Practices Related to Breast Cancer Screening Among Female Health Care Professionals: A Cross-sectional survey

Authors:

Humariya Heena (hmunshi@kfmc.med.sa)
Sajid Durrani (sdurrani@kfmc.med.sa)
Muhammad Riaz (medical.stn@gmail.com)
Isamme AlFayyad (ialfayyad@kfmc.med.sa)
Rabeena Tabasim (rtabasim@kfmc.med.sa)
Gazi Parvez (gchiken@kfmc.med.sa)
Amani Abu-Shaheen (aabushaheen@kfmc.med.sa)

Version: 4 Date: 17 Sep 2019

Author’s response to reviews:

1. The manuscript includes the list of authors and affiliations and the corresponding author, along with their email address.

2. Ethics approval and consent to participate section now clearly indicates that all participants provided informed consent for the study. The reference number for the ethics approval is included in this section.

3. In the Funding section, the role of the funding body in the design of the study; collection, analysis, and interpretation of data; and in writing the manuscript has been added.

4. Manuscript as a single, final, clean version has been uploaded. All relevant tables are also clean versions now.